Author's response to reviews

Title: Evaluation of an early step-down strategy from IV anidulafungin to oral azole therapy for the treatment of candidemia and other forms of invasive candidiasis: results from an open-label trial

Authors:

Jose Vazquez (jvazque1@hfhs.org)
Annette C. Reboli (reboli@rowan.edu)
Peter G. Pappas (pappas@uab.edu)
Thomas F. Patterson (Patterson@uthscsa.edu)
John Reinhardt (JReinhardt@johnfreinhardt.com)
Peter Chin-Hong (phong@php.ucsf.edu)
Ellis Tobin (Ehtmd1@pol.net)
Daniel H. Kett (dkett@med.miami.edu)
Pinaki Biswas (Pinaki.Biswas@pfizer.com)
Robert Swanson (swansr1@pfizer.com)

Version: 2 Date: 15 May 2013

Author's response to reviews:

Dear Editor

The trial registration number is now included as the last line of the abstract of the manuscript.

Yours faithfully
Robert Swanson